Tutorial #6 (for week 9 - March 5, 2004)

6th March 2004

Unfortunately, most of these questions are directly from previous year’s tutorials (instead of exams or NEW questions), so the answers are freely available here and there :( Q1: Code a “ToolTip” in Java/Swing.

Answer: Something like this:

Q2: Code a “ToolTip” for a Tk canvas, which puts up an informative box when you hover over any item on the canvas.

Answer: My copy of Tk has balloon/tooltips already, but different versions don’t. I hunted around and found some Tcl/Tk code at: http://www.multimania.com/droche/tkballoon/. The code tracks the mouse entering and leaving an item. Some versions of Tk allow you to just add a -tip “XXX” option to every widget. The following source code for a small program for balloons for Canvas widgets is based on Daniel Roche’s tkballoon code:

And the balloon routines are:

Q3: Give Java/Swing layout management code for the following:

Answer: Something like this:

Q4: The javax.swing.UIManager class is used to manipulate the look-and-feel of an application. How can you discover which look-and-feel strategies are implemented in the Java environment?

Answer: This was just an exercise in looking at the API. The class UIManager.LookAndFeelInfo can instantiate UIManager.LookAndFeelInfo objects. A method from UIManager class getInstalledLookAndFeels() returns all the look-and-feel strategies installed in the jdk being used. Thanks to Fong Qiyue for this clarification.

Q5: Write the Java/Swing code for a small application with two buttons, and a text entry box with a label, laid out as shown in this image.

Answer: Something like this: